

This is a peer-reviewed, post-print (final draft post-refereeing) version of the following published document:

Mohamed, A Fattaah, Singh, Satwinder, Irani, Zahir and Darwish, Tamer K ORCID logoORCID: https://orcid.org/0000-0003-1815-9338 (2013) An analysis of recruitment, training and retention practices in domestic and multinational enterprises in the country of Brunei Darussalam. International Journal of Human Resource Management, 24 (10). pp. 2054-2081. doi:10.1080/09585192.2012.723021

Official URL: http://www.tandfonline.com/doi/abs/10.1080/09585192.2012.723021 DOI: http://dx.doi.org/10.1080/09585192.2012.723021 EPrint URI: https://eprints.glos.ac.uk/id/eprint/2990

## Disclaimer

The University of Gloucestershire has obtained warranties from all depositors as to their title in the material deposited and as to their right to deposit such material.

The University of Gloucestershire makes no representation or warranties of commercial utility, title, or fitness for a particular purpose or any other warranty, express or implied in respect of any material deposited.

The University of Gloucestershire makes no representation that the use of the materials will not infringe any patent, copyright, trademark or other property or proprietary rights.

The University of Gloucestershire accepts no liability for any infringement of intellectual property rights in any material deposited but will remove such material from public view pending investigation in the event of an allegation of any such infringement.

PLEASE SCROLL DOWN FOR TEXT.



This is a peer-reviewed, post-print (final draft post-refereeing) version of the following published document:

Mohamed, A Fattaah and Singh, Satwinder and Irani, Zahir and Darwish, Tamer K (2013). *An analysis of recruitment, training and retention practices in domestic and multinational enterprises in the country of Brunei Darussalam.* The International journal of Human Resource Management, 24 (10), 2054-2081. ISSN 0958-5192

Published in The International journal of Human Resource Management, and available online at: <u>http://www.tandfonline.com/doi/abs/10.1080/0958519...</u>

We recommend you cite the published (post-print) version.

The URL for the published version is http://dx.doi.org/ 10.1080/09585192.2012.723021

## Disclaimer

The University of Gloucestershire has obtained warranties from all depositors as to their title in the material deposited and as to their right to deposit such material.

The University of Gloucestershire makes no representation or warranties of commercial utility, title, or fitness for a particular purpose or any other warranty, express or implied in respect of any material deposited.

The University of Gloucestershire makes no representation that the use of the materials will not infringe any patent, copyright, trademark or other property or proprietary rights.

The University of Gloucestershire accepts no liability for any infringement of intellectual property rights in any material deposited but will remove such material from public view pending investigation in the event of an allegation of any such infringement.

PLEASE SCROLL DOWN FOR TEXT

## An Analysis of Recruitment, Training and Retention Practices in Domestic and Multinational Enterprises in the country of Brunei Darussalam

Fattaah Mohamed, Satwinder Singh, Zahir Irani, Tamer Darwish Brunel University, Business School, UK.

# Abstract

This paper studies the behavioural differences in the recruitment, training and retention practices of domestic (DEs) versus multinational enterprises (MNEs) in the country of Brunei Darussalam. Hypotheses from literature survey predict MNEs to be more stringent in their recruitment and training and rigorous with promotion practices. Results show this is to be largely true. MNEs are found to be more rigorous in recruitment and place more emphasis on such traits as candidates' 'devotion to task', 'self-motivation', and 'independent judgment'. DEs rely more on internal appointments than external. MNEs place more emphasis on training; they also emphasise a stronger work culture by relying on 'induction by socialisation', and 'buddy system for mentoring'. When analysed by age, older firms were found to place more importance on language and commitment. They also rely on training via the buddy system and on external appointments for senior posts. Large firms place emphasis on employees' willingness to travel and work experience in other countries as the main recruitment criteria. Large firms also believe in external appointments for senior positions. The study which is one of few of its kind conducted in non-western environment, and the only one in the context of Brunei, adds to our understanding of HR practices in the context of two different genres of enterprises and has implications for future research.

**Keywords** – Recruitment, training, retention, domestic enterprises, multinational enterprises, t-test.

# 1. Introduction

A key factor in the success of any organisation is the way in which employees are recruited, trained, and retained within the organisation. This relationship, which is the cornerstone of sustaining competitive advantage vis-à-vis its rivals, has been studied in great theoretical detail by business economists (see, e.g., Baron, 1988; Milgrom, 1988; Appelbaum and Batt, 1994). This issue is now also being subjected to increasing empirical analysis, given its applied value to the corporate world and also for new research directions it may throw up. Koch and McGrath (1996), for example, found that careful recruitment and selection are positively related to the productivity of the labour force. Pfeffer (1998) also identified the importance of 'selective hiring' and has highlighted that a careful recruitment and selection process can ensure a better fit between organisational needs and employee skills and abilities. This is also echoed in the findings of Verburg (1998), who positively relates good selection processes with perceived profits, market share and investment. The empirical studies, some of which have been conducted in the context of multinational enterprises (MNEs), have

largely concentrated on comparing the human resources management (HRM) practices in the subsidiaries of MNEs in the context of developed economies (see e.g. Guest and Hoque, 1996; Boxall *et al.*, 2007). This paper fulfils a gap in the literature by conducting a study on similar lines in a non-western environment and in the context of an emerging economy. Furthermore, the study squarely compares the HR practices of recruitment, training, and retention (henceforth HR practices) in the context of domestic enterprises (DEs) and MNEs. As is now well recognised, MNEs have become key players in global commerce and are fast establishing themselves in developing economies<sup>1</sup>. This study, which is also first of its kind in the country context of Brunei, addresses four key related research questions. First, are the recruitment, training, and retention criteria much more detailed and stringent in MNEs vis-à-vis DEs? Second, do MNEs, given their global operations, ensure greater internal continuity by ensuring smooth executive succession and promotion? Third, how do the turnover rates of employees in the two types of enterprise (DEs and MNEs) compare? Finally, has the size and age of the firm any bearing on the operation of the three issues under investigation?

The paper is divided into the following sections. Section 2 summarises the literature on the three HR practices. Section 3 lists the hypotheses derived from the literature. Section 4 explains the research design, data, variables and methods. Section 5 presents the results. Section 6 includes an analysis on the extension of Basic Results on the Internal (Horizontal) Fit of HR Practices. The paper ends with a discussion, conclusions and synthesis.

## 2. Literature reference on HR practices

The HR practices explored in this paper are that of selection, recruitment, training, and internal opportunities. These practices were part of Pfeffer's (1994, 1998) original seven practices and have been measured by a number of researchers, largely in single country studies (e.g., Guest and Hoque, 1994; Delery and Doty, 1996; McDonnel *et al.*, 2011). The literature comparing these practices in a multi-country context is scant. The following summary brings together the work in this area together with the resulting hypotheses.

#### Recruitment and selection

<sup>&</sup>lt;sup>1</sup> At the last count there were 82053 parent corporations with 807363 foreign affiliates worldwide (440482 in non-developed world); the value of their outward foreign direct investment stock stood at US \$18.982 trillion (\$2.58 trillion in non-developed world) (World Investment Report 2009).

Recruitment and selection (staffing) is recognised as a major strategic international HRM (IHRM) practice that MNEs use to co-ordinate and control their global operations (Dowling and Schuler, 1990; Hendry, 1994) as well as to place the correct people in charge of these operations (Sparrow *et al.*, 1994). In terms of foreign operations though, recruitment and selection represents a challenge to MNEs, as international management teams combine and balance the knowledge of local market conditions and internal organisational competence. The importance of having the right people in the organisation at the right location globally at the right time can accelerate an organisation's international growth (Shen and Edwards, 2004).

Another key recruitment and selection issue is whether these organisations choose to recruit internally or externally. In the latter case, there are many options available to MNEs which include recruiting from the home country, the host country, a third country, or any combination of these (Scullion, 1995). Scholars such as Perlmutter (1969) and Dowling et al. (1999) have identified four approaches to international staffing: the ethnocentric approach (key positions filled by home country nationals); the polycentric approach (corporate headquarters run by home country nationals and 'foreign' subsidiaries run by host country nationals); the geocentric approach (best people for the key jobs, nationality not a consideration) and the regiocentric approach (where MNEs divide their operations into geographic sectors and divide staff between these sectors). Recruitment and selection has been found to differ across companies with different countries of origin; British firms, for example, seem to utilise external 'recruitment agencies' and 'executive search consultants' when filling managerial occupancies unlike, say, Taiwanese companies (Hsu and Leat, 2000). Another study by Wong and Birnbaum-Moore (1994) found that positions in Japanese and Swiss firms tended to be filled more by candidates from the home country than those from France, the Netherlands or the UK. Following on from our discussion thus far, the following two testable hypotheses can be listed:

- H1: Given the international operations of MNEs, the recruitment criteria in these enterprises would be more stringent than DEs.
- H2: Given that wider choices are available to MNEs, they rely more on external recruitment to fill vacancies than DEs.

The availability of internal career opportunities can decrease the chances of estrangement which can occur when new faces from outside fill senior positions (Noe *et al.*,

2006). Their availability also ensures that those moving to occupy the vacant position have insights into the organisation and position that they are occupying (Pfeffer, 1994). The provision of internal opportunities has also been found to be positively related to perceived profits, market share and investment (Verburg, 1998). In terms of cross-national or cross-cultural comparison studies, the results have been mixed. Ferner *et al.* (2001) discovered that German MNEs operating in Spain and the UK, exported German characteristics which included their long-term orientation in the companies. Von Glinow *et al.* (1999) found cross-cultural differences between Asian and European promotion criteria, stemming from differences in Hofstede's cultural values, which include a trait of high level of collectivism in Taiwan where the emphasis is on the importance of seniority compared to employee performance and experience in UK companies. This leads to the following hypothesis:

H3: Compared to DEs, the promotion criteria applied by MNEs is much more stringent.

## Training practices

The positive impact of this HR practice on employee motivation and performance has been widely reported (Harel and Tzafrir, 1999; Way, 2002; Winterton, 2007). The environment within which an organisation operates can affect the formal training systems in organisations. For example, it is common knowledge that training systems in the UK are largely market-led focusing on job-related skills, whereas in Germany training systems often take the shape of job-specific and general-purpose skills. This difference is echoed in a study by McGaughey and De Cieri (1999) and de Guzman *et al.*, (2011) where they reported that organisations are converging when it comes to macro-level variables, yet maintain a divergence in terms of micro-level variables. Scholars such as McPherson and Roche (1997) have also stressed the importance of training to MNEs, not only to its expatriates but also to host country nationals so as to develop international experience, thus increasing the quality of their international staff. In an empirical study, Taiwanese firms were found to conduct more on-the-job training programmes compared to their British counterparts (Cully, 1999).

A number of studies have related foreign ownership positively to the amount of training provided (Shen and Darby, 2006; Zheng *et al.*, 2006). Investment by MNEs into host countries would include investment in human capital in those countries in the form of training, in order to ensure a higher return on their investment (Lynch and Black, 1995). The

needs of international trade also force organisations to embrace higher standards to satisfy foreign markets, each different from one another, as well as challenges from unknown competitors. From this, H4 is derived as follows:

H4: MNEs place more emphasis on training compared to DEs.

## Retention practices

Companies' retention practices address the problem that comes about from employee turnover. An organisation's propensity to overcome problems stemming from employee turnover has to do with the retention incentives that can be utilised within the organisation as well as the organisation's drive to utilise them (Maertz and Campion, 1998). These incentives are also linked to other practices such as recruitment and selection, training and internal opportunities, as retention capacity is embedded within these practices (Reiche, 2008). For example, Shaw et al. (1998) highlight the importance of pay and benefits in encouraging employees stay within the organisation; Coff (1997) has added performance-based rewards to this, as well as a sense of possession in the form of profit sharing and stock ownership schemes. Magner et al. (1996) demonstrated that increased involvement in an organisation's decision-making processes can lead to a fall in turnover. In terms of multinational subsidiaries, Harzing and Sorge (2003) identified a tendency for subsidiaries to apply the HR practices utilised by their headquarters. This can bring about difficult challenges due to the different environments in which these organisations operate. Pressures from the home country combined with host country influences can affect the level of acceptance of transferred HR practices will receive. Therefore a clash between these two influences can affect the efficiency and effectiveness of HR practices utilised (Hofstede, 1998; Miller et al., 2001). Given these complexities, it is difficult, to state a priori what would be the effect of MNEs HR practices on employee turnover rate. The following hypothesis is based on the widely held belief in emerging transition economies that given the higher salaries and perks, the rate of employee turnover is lower in MNEs than DEs.

H5: MNEs have a lower turnover rate than DEs.

The size and age of a firm can also have an effect on the HR practices that organisations would choose to implement. Studies by various scholars (see Van Smoorenburg and Van der

Velden, 2000; Tan, 2001) all indicate significant differences in the HR practices implemented by organisations due to differences in their size. For instance, Matlay (1997) and Szamosi *et al.* (2004) have indicated that returns on training are unlikely to be realised by smaller firms due to their risk averse nature which encourages cost cutting and a lack of future planning. The following hypothesis is based on these findings:

H6: Firm size and age have a bearing on the HR practices of both DEs and MNEs.

## 3. Hypotheses

Following from the literature review outlined above, the six hypotheses given below are subjected to empirical testing in this paper:

- H1: Given the international operations of MNEs, the recruitment criteria in these companies are more stringent than DEs.
- H2: MNEs rely more on external recruitment to fill vacancy than DEs.
- H3: Compared to DEs, the promotion criteria applied by MNEs is much more stringent.
- H4: MNEs place more emphasis on training than DEs.
- H5: MNEs have a lower turnover rate than DEs.
- H6: Firm size and age have a bearing on the HR practices of both DEs and MNEs.

It needs to be emphasised that the above hypotheses are largely based on the literature review outlined above; they are also partly based on a general appreciation of MNEs' operations in host nations. For example, it is common knowledge in the international business literature that MNEs have three basic characteristics. These enterprises must be responsive to a number of essential environmental forces, including competitors, customers, suppliers, financial institutions, and governments. Second, given their global network, MNEs draw on a common pool of resources, including assets, patents, information, and human resources. And, third, MNEs are linked together by a common strategic vision (Rugman and Collinson, 2009). Their international activities are largely guided by these three principal characteristics. The practices analysed in this paper are no exception.

## 4. Research design, data, variables and methods

## Research design

The research design for this paper began with a review of the relevant literature as outlined above which led to the formulation of six testable hypotheses. The questionnaire instrument that was prepared, *inter alia*, to collect data for this paper went through a pilot study and a number of revisions before it was finally administered to the HR directors of DEs and MNEs operating in Brunei. Diagram 1 elucidates the process of designing research from start to finish.

## DIAGRAM 1 ABOUT HERE

## Data collection and sample

Data for this study comes from a primary survey administered to the HR directors of domestic and MNEs operating in the country Brunei Darussalam. Brunei is located in Southeast Asia bordering the South China Sea and Malaysia. A small country of less than half a million inhabitants, it obtained independence from Britain in 1984. The country is recognised by all nations and international organisations. Crude oil and natural gas production accounts for over half of national income. Around 63% (of 0.18m) of the workforce is engaged in industry, 33% in services, and 4% in agriculture. A number of large and small enterprises operate in the country. A count revealed a total of 465 such domestic and foreign enterprises. Of these, a sample size of 214 was selected<sup>2</sup>. In many instances, HR directors spent time with the authors to clarify questions before they completed the questionnaire. Cross checks reveal data to be consistent and reliable. Postal and face-to-face follow-ups yielded 151 usable replies amounting to a response rate of 70%. Having such a high response rate is noteworthy and meets the criteria suggested for applied research (Moser and Kalton, 1985). A number of similar studies comparing HR practices of organisations have also used similar methodology (see e.g., Bae et al., 1998; Hsu and Leat, 2000; Budhwar and Boyne, 2004). Of these 151 replies, 88 were from DEs (60%) and 63 were from multinational organisations (40%).

The questionnaire had eight sections that inquired about various aspects including the role of the HR director; recruitment, training, and retention; appraisals, incentives, and rewards; corporate culture; information on the domestic workforce; and company performance. This paper is based on the section on recruitment, training, and retention and data from first two sections on controls.

<sup>&</sup>lt;sup>2</sup> Based on the formula t o determine sample size,  $n = \frac{N}{1+N(e)^2}$  where N is the population size and e represents the confidence level.

## Construction of variables

Recruitment variables used in this paper asked HR directors about the importance that was placed on qualifications and personal characteristics (7 questions each on a Likert scale of 1-5) and the importance of internal and external recruitment to various posts (3 questions) ( see Table-1 for details). Six questions were asked on managing senior executive succession (Table-2). The training methods section asked six questions (Table-3). Questions on performance criteria explored eight questions (Table-4). The age and size of organisations were used as control variables. Older firms were defined as those established for 15 years or more. Following European Commission guidelines (2003), firms with more than 50 employees was defined as of large size.

## Methods

The sample replies were demarcated into DEs and MNEs. The term MNEs has been defined broadly to include all companies with foreign ownership, majority or minority including joint ventures (JVs). JVs with varying degrees of foreign equity participation are an essential means of gaining entry, as Brunei does not always permit 100% foreign equity ownership. However, as is well known, even with minority ownership, a degree of control over an enterprise can be exercised via clauses in management contracts and the terms of the joint venture. As a result, when conducting an ANOVA test, as expected, we found the two groups (MNEs and JVs with foreign equity participation) to have similar characteristics, permitting us to classify them in one group.

Before undertaking empirical work, data was screened for any inaccuracies. The effects of any missing data, outliers, and assumptions underlying multivariate techniques were assessed and data was found to be consistent for subjecting to statistical analysis. Research questions emanating from the literature review and leading to testable hypotheses required that, through the use of a statistical methodology, we may be able to clearly discern the differences between the recruitment, training, and retention practices of DEs and MNEs. Thu, it was decided that the t-test (described below) be adopted for each of the practices, and also to verify the Internal (Horizontal) Fit of HR Practices. However, before we could do this, we had to apply Levene's test to ascertain whether the variances were different for the different

groups. Levene's criterion (Levene, 1960) tests the null hypothesis that the difference in variances in different groups is zero. The test works on the deviation scores, which is the absolute difference between each score and the mean of the group where it came from (Glass, 1966). The formula for Levene's test can be defined as follows:

$$W = \frac{(N-k)}{(k-1)} \frac{\sum_{i=1}^{k} N_i (Z_{i.} - Z_{..})^2}{\sum_{i=1}^{k} \sum_{j=1}^{N_i} (Z_{ij} - Z_{i.})^{2'}}$$

Where W is the result of the test and k represents the number of different groups to which the samples belong. N is the total number of samples and  $N_i$  is the number of samples in the *i*th group.  $Y_{ij}$  is the value of the *j*th sample from the *i*th group.

$$Z_{ij} = \begin{cases} \begin{vmatrix} Y_{ij} - \bar{Y}_{i.} \\ | Y_{ij} - \bar{Y}_{i.} \end{vmatrix} & \bar{Y}_{i.} \text{ is a mean of the ith group whilst } \tilde{Y}_{i.} \text{ is a median of the ith group.} \\ Z_{.} \text{ is the mean of all } Z_{ij} & (\text{i.e. } Z_{..} = \frac{1}{N} \sum_{i=1}^{k} \sum_{j=1}^{N_{i}} Z_{ij}) \text{ and } Z_{i.} \text{ is the mean of the } Z_{ij} \text{ for group } i \\ (\text{i.e. } Z_{i.} = \frac{1}{N} \sum_{j=1}^{N_{i}} Z_{ij}). \text{ If Levene's test is significant at } p \leq 0.05, \text{ we can conclude that the null hypotheses are incorrect and that the variances are significantly different where the assumptions of homogeneity of variances are violated. If the test is insignificant ( $p > 0.05$ ), the variances are roughly equal and the assumption is tenable. The results of Levene's test allows for adjustments to be made in subsequent tests when variances in experimental groups are found to be unequal. The independent t-test looks at the differences between groups and the formula arrived at to conduct it. It also takes into account the unequal number of participants in each group (in our case, DEs vs. MNEs) as the variance of each sample can be weighted. The *t* statistic can be stated as: =  $\frac{\bar{X}_1 - \bar{X}_2}{\sqrt{\frac{s_1^2}{N_1 + \frac{s_2^2}{N_2}}}}$ .$$

#### 5. Results

#### Recruitment issues

Table-1 lists the level of importance HR managers placed on qualifications and personal traits in recruiting new employees. The table also lists the importance of internal and external recruitment to various posts in the company. The Mean column shows that, on average, MNEs placed higher emphasis on all the attributes of personal qualifications. The higher mean for MNEs is also statistically significant for all seven attributes but one—'wide range of experience'. MNEs seem to place particular emphasis on 'language' (Md = -0.502, t = -3.574, p < 0.000), followed by experience in similar jobs and professional qualifications. With regard to personal characteristics, the mean differences for all the attributes are markedly higher for MNEs and these higher mean differences are also statistically significant at .01 levels. It seems that MNEs place markedly more emphasis on a candidate's willingness to travel, devotion to task, self motivation, potential to grow with the job, independent judgment, commitment to the company, and willingness to learn. These results support H1 which stated that the recruitment criteria would be more stringent in MNEs than DEs.

## TABLE-1 ABOUT HERE

The last three rows of Table-1 report the relative importance of internal and external recruitment to various posts in the company. The Likert scale ranged from 1 - 5 (1 being entirely internal and 5 entirely external). Here the DEs lead the mean score over MNEs when it comes to senior manager (reporting to chief executive) appointments, and junior manager (supervising operatives) appointments. The higher mean for these two attributes tells us that when it comes to such appointments, DEs rely more on internal appointments than external, and that this reliance is statistically significantly higher for DEs. No statistically significant difference is seen when it comes to the appointment of a professional specialist.

Table- 2 presents the responses to the question 'please indicate your views on managing senior executive succession?' The response was on a 1-5 scale with 5 indicating total agreement. The striking result in this table is the opposite views held by two genres of enterprises. MNEs seem to believe in open market appointments (Md = 0.886, t = 5.216, p < 0.00) and DEs in internal appointments (Md = -0.429, t = -2.148, p < 0.01). The significant reliance of DEs on non-executive directors to play a dominant role in selecting executive successors also supports this result which confirms H2.

#### TABLE-2 ABOUT HERE

## Training Issues

Table 3 presents the response to question 'what are the most applicable methods of training new employees in junior management?' The responses were coded on a 1-5 scale with 5 as the most applicable method. Three methods stand out on which MNEs rely significantly more than DEs. These are: 'buddy system mentoring' (Md = -0.560, t = -3.104, p < 0.00), 'work placement with partners' (Md = -0.472, t = -2.572, p < 0.00) and 'induction by

socialisation' (Md = -0.448, t = -2.268, p < 0.01). These results show support for H3, and stronger support for H4 in that MNEs place more emphasis on training as a whole which has a higher mean on all methods. MNEs also emphasise a stronger work culture as shown by the significant statistical differences in 'induction by socialisation', 'work placement with partners' as well as the 'buddy system / mentoring'.

## TABLE-3 ABOUT HERE

## Internal Career Opportunities

Table 4 reports the results of the question posed to HR directors 'please indicate the main criteria of individual or group performance used in assessing cases for promotion?' Eight alternatives were offered on a 1-5 scale with 5 standing for most applicable. Interestingly, looking at the mean responses and those that are statistically significant, there does not seem to be a great deal of difference in responses from MNEs and DEs, save for marginally more emphasis that MNEs pay to technical skills (Md = -0.299, t = -1.959, p < 0.05) 'keeping within budget' (Md = -0.209, t = -1.452, p > 0.05), and 'professionalism' (Md = -0.188, t = -1.433, p > 0.05). This lends only partial support for H5 that stated that MNEs will employ much more stringent criteria for promotion.

## TABLE-4 ABOUT HERE

#### HR-related outcomes: Retention / Turnover rate

A straightforward question on the turnover rate of employees was posed to understand the attrition rate and if it was higher or lower for MNEs vs. DEs. We took an attrition rate of 1-3% to be on the lower side and compared results for the two genres of enterprise. The results reported in Table-5 indicate that the attrition rate is significantly lower for MNEs than DEs (Md = 1.084, t = 3.768, p < 0.05). This result is totally in tune with H6.

#### TABLE-5 ABOUT HERE

Size and age of the enterprise

The results presented thus far do not account for the size and age of the firm. It is probable that the significant results may cease to be significant when the sample firms are divided into groups of different sizes. Similarly, it is possible that older firms have a different work culture and treat their employees differently. It was thus decided to analyse the significant results with sample firms divided into two group sizes and ages. Table-7 present an analysis based on two sample sizes: small (firms with up to 50 employees) and large (firms with more than 50 employees). Table-8 examines the results with sample sizes divided into two age groups: one group of firms who have been established during the past 14 years and the second group comprising firms established for over 15 years. The results are remarkable in that significant results stay significant for all the young firms and only for selected traits for older firms. For example, for older firms, the importance of language (Md=-0.429, t=2.32, p < .05); and commitment (Md=-0.484, t=3.59, p < .00) stand out. Similarly, training via the buddy system (Md=-0.474, t=2.08, p<.05); and external appointments for senior posts' selection (Md=-0.72, t=3.51, p<.00) stand out. With regard to the size of the firms, the previous results largely seem to hold much more for smaller firms than larger firms. Larger firms seem to place emphasis on employees' willingness to travel (Md=-0.858, t=-3.245, p < .00; and work experience in other countries (Md=-0.67, t=2.80, p < .00) as the main recruitment criteria. Large firms also believe in external appointments for senior positions (Md=0.918, t=3.66, *p*<.00).

## TABLE- 6 ABOUT HERE

## TABLE-7 ABOUT HERE

#### 6. Extension of Basic Results: the Internal (Horizontal) Fit of HR Practices

Having only considered, analysed and compared the HR practices at an individual or item level in the previous analysis, we now assess whether the HR practices are integrated and coherent as a whole in DEs and MNEs in the context studied. For us to be able to do so, we first deal with HR practices as composite variables, where each practice reflects all its related items. We then assess the internal or horizontal fit between HR practices to examine whether they work together as an integrated bundle or system of practices. Importantly, some researchers have assessed the bundles or synergies (the internal or horizontal fit) of the whole HR practices in HRM studies. In recent literature, the internal consistency or bundling of HR practices has been one of the core theoretical concerns in HRM. The discussion centres on

whether HR practices should be integrated or incorporated as a coherent system of practices that are mutually supportive (Delery, 1998), which can be best represented by bundling the HR practices. However, the measurement is still a matter of debate amongst HRM researchers (Guest, 2011; Singh et al., 2012). It is commonly assumed that HR bundles or complementarities must be more than simply the additive sum of each practice's independent effects (Macky & Boxall, 2007). Instead, the concept of the internal fit of HR practices implies that such practices must have a synergistic or a mutual association which can be statistically reflected by the interaction terms test (Huselid, 1995; Wood, 1999; Macky & Boxall, 2007). Therefore, HRM scholars who have investigated such arguments consider the interaction terms amongst HR practices as the best indicator of HR practices as a coherent system. As suggested by researchers who have examined the internal fit of HR practices (e.g., Huselid, 1995; Delaney & Huselid, 1996; Macky & Boxall, 2007; Wood & Menezes, 2008), we examine the potential of complementarities or the internal fit in terms of interactive relationship amongst HR practices. We test the two-way interaction terms amongst HR practices for both DEs and MNEs separately with the aim of exploring in which set of enterprises HR practices are more or better integrated and coherent as whole.

Table 8 first lists the overall level of importance HR managers placed on the HR practices of recruitment and selection, training, and internal career opportunities. The Mean column shows that MNEs placed higher emphasis on all these HR practices. Markedly, the higher mean for MNEs is also statistically significant for the HR practices of recruitment and selection (Md = -4.256, t = -2.621, p < 0.01), and training (Md = -1.630, t = -2.621, p < 0.01); however, the difference is not significant for the HR practice of internal career opportunities. The second part of Table 8 also highlights the internal fit amongst the HR practices. Our results show that HR practices can be better integrated, and are coherent as a bundle or system of practices in MNEs compared to DEs. All results in Table 8 are statistically significant in support of better internal consistency in MNEs.

As stated earlier, the idea of internal consistency amongst HR practices is one of the core theoretical concerns in HRM. However, little progress has been made in this regard (Guest, 2011). Notably, HRM researchers believe that such powerful and logical combinations can have great impacts on the bottom-line issue within the organisation. Therefore, future comparative HRM works could retain a focus on this issue and also go further by developing some form of links between different bundles of HR practices with different business

strategies (examining the external or vertical fit). Some researchers have suggested that organisations may have different bundles of practices that can perfectly work with different types of business strategies, which, at the end, can lead companies to perform better (Delery & Doty, 1996; Razouk, 2011). We recommend that future comparative HRM work can go further than simply comparing individual HR practices, and instead conduct comparisons in terms of the bundles of practices, and also link such practices with business strategies, and assess the impacts of these bundles on organisational performance through comparative means. For instance, some HRM researchers (e.g., Youndt *et al.*, 1996; Hoque, 1999) have found that the positive relationship between HRM practices and the overall performance of the companies is dependent on the fit between HR practices and the business strategies in terms of investigating their association with the studied HR practices. This could be an area of investigation in future studies.

#### **TABLE-8 ABOUT HERE**

#### 7. Discussion, Synthesis and Conclusions

The purpose of this paper was to study the behavioural differences in the recruitment, training and retention practices of domestic versus multinational enterprises in the country of Brunei Darussalam. There is scant literature that compares the comparative aspects of these practices and none so in the country context of Brunei. Based on a survey of available literature and generally held beliefs, hypotheses were developed whereby MNEs were predicted to be much more stringent in their recruitment and training practices of new recruits and rigorous with promotion practices. On balance we found this to be true. In addition, it also transpires that younger firms, established in last decade or so, and smaller firms, are likely to carry out these practices more rigorously than larger and older established firms. These results, without ambiguity, confirm the importance of the three HR practices for MNEs examined in this paper. The findings highlight the state of development as well as the nature of either individual organisations or the country as a whole (Rowley and Benson, 2002). Some of the results are better illustrated when we look at the HR practices individually. The differences seen between DEs and MNEs with regard to the recruitment and selection process can be due to differences between these enterprises, and possibly attributable to the nature of globalisation. This echoes results obtained by Myloni et al. (2004) who found selection

methods in domestic companies lagging behind multinationals in Greece. These differences are also reflected in the findings of Huang (2000), albeit in a more eastern versus western context. His results showed distinct contrasts in most HR practices including staffing, training, and career path design, between western MNEs and companies based in the East including domestic Taiwanese companies. In the case of the present study, the finding that DEs lag behind MNEs could also relate to an interesting implication. In the HRM literature, researchers have mostly focused either on the universalistic or the contingency approaches when they study the HR practices. The universalistic approach has an ideal group of 'best practices' that are supposed to continuously generate superior organisational performanceregardless of the circumstances and the industry-whilst the contingency theory states that HR practices will be always dependent on the environment, context, or the circumstances surrounding the organisation, and that they can be changed upon. Notably, our results give more support to the 'contingency approach' than the 'best practices' approach. We can see that the same HR practices in DEs are different and lag behind those found in MNEs. This difference may be attributed to the effects of the different non-Western cultural context, as well as circumstances surrounding the DEs in the country of Brunei. Such results tell us that we cannot only consider the studied HR practices as best practices, regardless of the environment or the industry, as stated by some scholars (e.g., Huselid, 1995; Pfeffer, 1998).

Additionally, there could be other reasons why the results of this paper support MNEs' focus on internal recruitment when compared with domestic organisations. One reason could be the quality of the domestic workforce. MNEs may not rate the quality of the domestic workforce very highly; internal recruitment could also mean the placement of expatriates into higher positions in the organisation to smoothly oversee operations. The use of expatriates by MNEs in this case are in tune with the findings of Shen and Edwards (2004), who found that expats were used to control the coordination of international activities and oversee subsidiary operations and administrative and financial aspects of the organisation. Another reason for the disparity between domestic companies and MNEs is the culture of the host country. If the host country ranks highly on the collectivism scale, these cultures may prefer to use the internal labour market as it promotes loyalty to the firm (Budhwar and Khatri, 2001). As internal recruitment is susceptible to cultural differences, MNEs may have to adjust towards the domestic environment, as is also evident in the findings of Myloni *et al.* (2004), with the result that DEs rely heavily on the internal labour market and MNEs on the external labour market.

The results obtained in this paper also found support for the proposed hypotheses on training methods and although the statistical differences on its importance is higher for MNEs, the mean score emphasising the importance of training for DEs comes out to be high as well. This shows that both sets of enterprises rank the importance of training their employees highly. Informal training methods, as opposed to formal methods (Wright et al., 2002), seem to be popular with both types of enterprise. This is in line with the findings of convergence hypotheses (Myloni et al., 2004; Carr and Pudelko, 2006; Ayndinli, 2010). They are also indirectly in tune with the findings of Zheng et al. (2007) which showed a concern about the relevance and quality of training programmes administered externally. It seems that the benefits that training can bring to organisational performance, especially in Asia (see Osman-Gani and Jacobs, 2005; Jaw et al., 2006; Bao and Analoui, 2011), have increased the adoption of popular, informal training methods. This supports arguments by Von Glinlow et al. (2002) and Schuler and Jackson (2005) whereby the best HRM practices tend to be shared and are used globally. The reasons for the mean results on training methods being statistically significantly higher for MNEs could also reflect the defeatist culture sometimes inherent in small host countries. DEs may find competition with MNEs too intense to devote effort to train their labour only for it to be poached by MNEs. In a study, Lawler et al. (1995) observed that Asian firms find employee training to be risky and costly as there are no immediate returns and these trained employees may leave before any benefits can be realised.

These results obtained showed partial support for the hypothesis on internal career opportunities leading to executive succession. Unlike the criteria for recruitment, this issue has not been deeply explored in HRM literature. However, its importance has been highlighted by various authors such as Delery and Doty (1996) and Joseph and Dai (2009) who found positive connections between internal opportunities and organisational performance measures. Here, we found that MNEs prefer to avoid competition between internal candidates by 'preparing' one person well in advance, probably to maintain harmony and avoid disruption which can impact on performance. Results also showed that DEs, on the other hand, preferred external appointments for executive succession which could bring new ideas into the system. Compared to DEs, MNEs reliance on sound individual technical skills for promotion purposes displays their preference for technically sound employees in senior positions. This is probably due to the pressure of global business in which employees with sound technical skills (particularly in manufacturing concerns) tend to perform better than

employees lacking such skills. One issue that we did not set out to test but which we believe is worth touching upon here is the recent interest in 'talent management' in enterprises (Vaiman & Vance, 2008; Inskeep & Hall, 2008). In this paper, we have examined three essential components of talent management, which also include additional strands, such as replacement planning, leadership development, and mentoring. Such issues, grounded in the strategic human resource management literature, are intended to engage employees to retain them in the company, and can be key elements in leveraging competitive advantage (Gratton, 2000; Becker, Huselid & Ulrich, 2001). Our results in this paper show that MNEs are more strategic than DEs in terms of recruitment, training, and retention issues. Their implicit emphasis on these issues might have the undercurrent of effective talent management with a view to keep them in their organisations. Indirect evidence of this was also found in this paper in results that showed statistically significant findings on the lower rate of turnover of employees for MNEs. This is consistent with the theoretical and empirical work to date on HRM as researchers have proven that effective HR practices could lead to a lower turnover rate (e.g., Huselid, 1995; Guthrie et al., 2009). The results are not in tandem with Wood et al.'s (2006) results which showed no relationship between the practices and employee turnover.

#### Contribution

This paper highlighted the gap in the literature on comparative studies of HR practices between DEs and MNEs, particularly in the context of small, emerging economies. The examination of these HR practices has been conducted looking at the practices as individual groups and not as one block to avoid obscuring any important differences (Myloni *et al.*, 2004). It is likely that these differences in HR practices come about due to differences in national culture, external organisational factors such as regulation by government institutions, internal organisational factors such as organisation size and ownership type, as well as the national view of HR within a country which can encompass historical competence, the role of HR and its development (Sparrow and Hilltrop, 1994). The support of two hypotheses proposed in this paper shows a divergent view when it comes to the importance of HR in DEs and MNEs, with the HR systems of DEs still lagging behind the more developed systems of the MNEs. As the comparisons made in this paper of the recruitment and selection, training and internal career opportunities in domestic and multinational organisations have shown, there is a clear difference in the practices adopted by these two genres of enterprise. Previous

studies conducting comparisons on the HR practices in domestic and multinational companies have obtained varying results depending on the focus of the study as well as the context in which they have been conducted. Studies looking at labour relations have yielded both convergent and divergent results. Studies in developing or newly developed countries have also found divergent results (Myloni et al., 2004; Chen et al., 2005). However, the findings of this paper should not be taken as a case of 'reverse diffusion' (Hayden and Edwards, 2001), in which MNEs perceive their home systems as inferior to the host country standards and would be unable to transfer effectively to their subsidiaries. This difference echoes the findings of many authors comparing the HRM practices in DEs and MNEs such as Papalexandris (1987) who found more sophisticated HR practices in foreign companies directed by their overseas headquarters; and Purcell (1987) who found MNEs focusing on getting higher commitment from their employees. Yuen and Kee (1993) saw more extensive and formal HR practices used by US multinationals in their comparisons. Authors such as Geary and Roche (2001) have also identified differences in HR practices when comparing DEs and MNEs, be it through the dominance of 'country of origin' over 'host country effects' or changes in host country institutional policies and practices. The findings of this paper contrast with those of Guest and Hoque (1996), Myloni et al. (2004) and Chen et al. (2005) who finds similarities in the HR systems used by DEs and MNEs, finding convergence due to the 'culture sensitive view'.

## Limitations and avenues for further research

In spite of the contributions this study makes to the literature, researchers acknowledge some of its limitations. By using a sole respondent, that is the HR director from each organisation, the present study could suffer in respect of the accuracy of their perceptions. Though it is extensively utilised in research, the 'key informant approach' can produce problems in the form of common method variance. Resources permitting, this could be improved upon by aiming for other managers to check the accuracy of their opinions. Fortunately, in this case, tests were performed to check for the presence of common method variance and these tests showed that this would not be a major concern. Second, this study looks at the individual HR practices involved in 'work flow' (recruitment and selection, training and development, internal career opportunities). Although looking at these practices individually has been done in previous studies (e.g. Chen *et al.*, 2005; Myloni *et al.*, 2004), we are beginning to identify linkages between the individual practices explored in this paper. These linkages could also

extend to other HR practices such as incentives and rewards and performance appraisals as well as the performance indicators of these enterprises. Further exploration of these linkages should be done in the future to add to the existing body of literature. There is also a scope for future research into the types of HR practices relevant for the different sectors and industries so as to determine the effectiveness of HR practices in different sectors. Our analysis is at the aggregate level. An additional number of HR practices can also be added to those analysed in this paper.

## Implications

This paper recognises the benefits associated with effective and efficient recruitment and selection practices and policies utilised by organisations when selecting the right employees for the jobs; these are critical to their success. The misuse of this practice can lead to undesirable HRM outcomes, such as demoralised and demotivated employees, which are costly to all organisations. Further, during the final stage of the recruitment and selection process, HR directors should consider candidates' potential, whereby qualifications, characteristics and work-related values and attitudes would most closely fit into the requirements of the offered positions, as well as those who can fit well with the organisation's needs and culture. This can effectively ensure that required tasks are carried out as expected, and harmony with colleagues is established. Our results show that MNEs in Brunei face lower turnover rates than DEs operating in the same context, and that MNEs have better HR practices than DEs. These practices include more stringent recruitment and selection policies, better training systems, and more internal career opportunities. Taking a cue from this, organisations are therefore able to reduce employee turnover at the start of an employee's career by ensuring a good fit between individual and organisational culture at the recruitment stage through more stringent processes, such as focusing on the quality of their qualifications. Likewise, extensive formal training can lower the percentage of employees leaving each year. Additionally, organisations that promote from within and actively prepare their employees for higher positions would improve employee turnover rates which organisations face annually.

<u>Acknowledgements</u>: Authors are indebted to an anonymous referee and to the editor of this journal for providing valuable feedback on an earlier draft of the paper which led to several

useful alterations in the paper. All shortcomings are ours.

## References

- Appelbaum, E., and R. Batt. (1994), *The New American Workplace: Transforming Work Systems in the United States*, New York: ILR Press.
- Aydinli, F., (2010), 'Converging human resource management: A comparative analysis of Hungary and Turkey', *International Journal of Human Resource Management*, 21, 9, 1490-1511.
- Bae, J., Chen, S., and Lawler, J. (1998), 'Variations in human resource management in Asian countries: MNC home-country, and host-country effects', *International Journal of Human Resource Management*, 9, 4, 653–70.
- Bao, C., and Analoui, F. (2011), 'An exploration of the impact of strategic international human resource management on firm performance: The case of foreign MNCs in China', *International Journal of Management and Information Systems*, 15, 4, 31-40.
- Baron, J. (1988), 'The employment relation as a social relation', *Journal of Japanese and International Economies*, 2, 492-525.
- Boxall, P., Purcell, J., and Wright, P. (2007), *The Oxford Handbook of Human Resource Management*, New York: Oxford University Press.
- Budhwar, P. and Boyne, G. (2004), 'Human resource management in the Indian public and private sectors: An empirical comparison', *International Journal of Human Resource Management*, 15, 2, 346-370.
- Budhwar, P., and Khatri, N. (2001), 'HRM in context: Applicability of HRM models in India', *International Journal of Cross Cultural Management*, 1, 3, 333–56.
- Carr, C. and Pudelko, M. (2006), 'Convergence of management practices in strategy, finance and HRM between the USA, Japan and Germany', *International Journal of Cross Cultural Management*, 6, 1, 75-99.
- Chen, S., Lawler, J., and Bae, J. (2005), 'Convergence in human resource systems: A comparison of locally owned and MNC subsidiaries in Taiwan', *Human Resource Management*, 44, 237-256.
- Coff, R.W. (1997), 'Human assets and management dilemmas: Coping with hazards on the road to resource-based theory', *Academy of Management Review*, 22, 2, 374–402.
- Cully, M. (1999), Britain at Work: As Depicted by the 1998 Workplace Employee Relations Survey, Routledge, London.
- de Guzman, G.M., Neelankavil, J.P., and Sengupta, K, (2011), 'Human resource roles: Ideal versus practiced: A cross-country comparison among organisations in Asia', *International Journal of Human Resource Management*, 22, 13, 2665-2682.
- Delaney, J.T. and Huselid, M.A. (1996), 'The impact of human resource management practices on perceptions of organizational performance', *Academy of Management Journal*, 39, 4, 949-969.
- Delery, J. (1998), 'Issue in fit in strategic human resource management: Implications for research', *Human Resource Management Review*, 8, 3, 289-309.
- Delery, J. E., and Doty, D. H. (1996), 'Modes of theorizing in strategic human resource management: Tests of universalistic, contingency', *Academy of Management Journal*, 39, 4, 802–835.
- Dowling, P.J. and Schuler, R.S. (1990), International Dimensions of Human Resource Management, Boston, MA: PWS-Kent.
- Dowling, P.J., Schuler, R.S. and Welch, D. (1999), International Human Resource Management: Managing People in a Multinational Context, Cincinnati, OH: South-

Western College Publishing.

- European Commission (2003), 'Commission recommendation of 6 May 2003 concerning the definition of micro, small and medium-sized enterprises', *Official Journal of the European Union*, 124, 37.
- Ferner, A., Quintanilla, J. and Varul, M.Z. (2001), 'Country-of-origin effects, host country effects, and the management of HR in multinationals: German companies in Britain and Spain', *Journal of World Business*, 36, 2, 107-127.
- Geary, J. F., and Roche, W. K. (2001), 'Multinationals and human resource practices in Ireland: A rejection of the 'new conformance thesis', *International Journal of Human Resource Management*, 12, 1, 109-127.
- Guest, D., and Hoque, K. (1996), 'National ownership and HR practices in UK greenfield sites', *Human Resource Management Journal*, 6, 4, 50-74.
- Guest, D.E., (2011), 'Human resource management and performance: Still searching for some answers', *Human Resource Management Journal*, 22, 1, 3 13.
- Guest, D.E., and Hoque, K. (1994), 'The good, the bad and the ugly: Employee relations in new non-union workplaces', *Human Resource Management Journal*, 5, 1–14.
- Guthrie, J., Flood, P., Liu, W., and MacCurtain, S., (2009), 'High performance work system in Ireland: Human resource and organisational outcomes', *The International Journal of Human Resource Management*, 20, 1, 112-125.
- Harel, G.H and Tzafrir, S.S. (1999), 'The effect of human resource management practices on the perceptions of organizational and market performance of the firm', *Human Resource Management*, 3, 38, 185–200.
- Harzing, A.W. and Sorge, A. (2003), 'The relative impact of country of origin and universal contingencies in internationalization strategies and corporate control in multination enterprises: Worldwide and European perspectives', *Organization Studies*, 24, 187–214.
- Hayden, A. and Edwards, T. (2001), 'The erosion of the country of origin effect; A case study of a Swedish multinational company', *Industrial Relations*, 56, 116-140.
- Hendry, C. (1994), Human Resource Strategies for International Growth, London: Routledge.
- Hofstede, G. (1998), 'Think locally, act globally: Cultural constraints in personnel management', *Management International Review*, 38, 7–26.
- Hoque, K. (1999), 'Human resource management and performance in the UK hotel industry', *British Journal of Industrial Relations*, 37, 3, 419–443.
- Hsu, Y. and Leat, M. (2000), 'A study of HRM and recruitment and selection policies and practices in Taiwan', *International Journal of Human Resource Management*, 11, 2, 413–35.
- Huang, T. (2000), 'Human resource management practices at subsidiaries of multinational corporations and local firms in Taiwan', *International Journal of Selection and Assessment*, 8, 1, 22-8.
- Huselid, M. (1995), 'The impact of human resource management practices on turnover, productivity, and corporate financial performance', *Academy of Management Journal*, 38, 3, 635–670.
- Inskeep, Nancy A and Hall, Bettie. (2008), 'Rewards and recognition concepts that support talent and knowledge management initiatives', In Vaiman and Vance (Ed.), *Smart talent management*, pp161-175. Cheltenham, Uk: Edward Elgar.
- Jaw, B.S., Wang, C.Y.P. and Chen, Y.H. (2006), 'Knowledge flows and performance of multinational subsidiaries: The perspective of human capital', *International Journal of Human Resource Management*, 17, 2, 225-44.
- Joseph, K.E. and Dai, C. (2009), 'HRM practices and organizational performance: An empirical analysis', *International Journal of Business and Management*, 4, 8, 117-127.

- Koch, M.J., and McGrath, R.G. (1996), 'Improving labor productivity: Human resource management policies do matter', *Strategic Management Journal*, 17, 335–354.
- Lawler, J., Jain, H., Venkata Ratnam, C. and Atmiyananda, V. (1995), 'Human resource management in developing economies: A comparison of India and Thailand', *International Journal of Human Resource Management*, 6, 319-46.
- Levene, H. (1960), 'In contribution to probability and statistics' in *Essays in honour of Harold Hotelling*, ed. Olkin et. al., Stanford University Press, pp.278-292.
- Lynch, L.M. and Black, S.E. (1995), 'Beyond the incidence of training; Evidence from a national employer survey', NBER Working Paper 5231, National Bureau of Economic Research, New York, NY.
- Macky, K., and Boxall, P. (2007), 'The relationship between 'high-performance work practices' and employee attitudes: An investigation of additive and interaction effects', *The International Journal of Human Resource Management*, 18, 4, 537 -567.
- Maertz, C.P. and Campion, M.A. (1998), '25 years of voluntary turnover research: A review and critique', In Cooper C.L. and Robertson I.T. (Eds.), *International review of industrial and organizational psychology*, 13, 49–81, Wiley, NY.
- Magner, N., Welker, R.B. and Johnson, G.G. (1996), 'The interactive effects of participation and outcome favourability on turnover intentions and evaluations of supervisors', *Journal of Occupational and Organizational Psychology*, 69, 2, 135–143.
- Matlay, H. (1997), 'The paradox of training in the small business sector of the British economy', *Journal of Vocational Education and Training*, 49, 4, 573-88.
- McDonnel, A., Stanton, P., and Burgess, J. (2011), 'Multinational enterprises in Australia: Two decades of international human resource management reviewed', *Asia Pacific Journal of Human Resources*, 49, 1, 9-35.
- McGaughey, S.L., and De Cieri, H. (1999), 'Reassessment of convergence and divergence dynamics: Implications for international HRM', *International Journal of Human Resource Management*, 10, 2, 235-250.
- McPherson, A.H., Roche, W.K. (1997), 'Peripheral location equals localised labour? Multinationals and the internationalization of training and development in Ireland', *International Journal of Human Resource Management*, 8, 4, 369-84
- Milgrom, P. (1988), 'Employment contracts, influence activities, and efficient organization design', *Journal of Political Economy*, 96, 42-60.
- Miller, J.S., Hom, P.W., and Gomez-Mejia, L.R. (2001), 'The high cost of low wages: Does Maquiladora compensation reduce turnover?', *Journal of International Business Studies*, 32, 3, 585–595
- Moser, C.A. and Kalton, G. (1985), Survey Methods in Social Investigation, Gower: Aldershot.
- Myloni, B., Harzing, A., K. and Mirza, H. (2004), 'Human resource management in Greece: Have the colours of culture faded away?', *International Journal of Cross Cultural Mangement*, 4, 1, 59-76.
- Noe, J., Hollenbeck, B. Gerhart and P. Wright. (2006), *Human Resources Management: Gaining A Competitive Advantage*, Mc- Graw-Hill Co.NY.
- Osman-Gani, A.M., and Jacobs, R.L. (2005), 'Technological change and human resource development practices in Asia: A study of Singapore-based companies', *International Journal of Training and Development*, 9, 4, 271-80.
- Papalexandris, N. (1987), 'Factors affecting management staffing and development: The case of Greek firms', *European Management Journal*, 6, 1, 67–72.
- Pfeffer, J. (1994), Competitive advantage through people, Harvard Business School Press, Boston.

- Pfeffer, J. (1998), 'Seven practices of successful organizations', *California Management Review*, 40, 2, 96–124.
- Purcell, J. (1987) 'Mapping management styles in employee relations', Journal of Management Studies, 24, 5, 533-548.
- Razouk, A. (2011), 'High performance work systems and performance of French small- and medium-sized enterprises: Examining causal order', *The international Journal of Human Resource Management*, 22, 2, 311-330.
- Reiche, B. S. (2008), 'The configuration of employee retention practices in multinational corporations' foreign subsidiaries', *International Business Review*, 17, 676-687
- Rowley, C., and Benson, J. (2002), 'Convergence and divergence in Asian HRM', *California Management Review*, 44, 2, 90-109
- Rugman, A.M., and Collinson. S. (2009), *International business*. 4th ed. London: FT Prentice Hall.
- Schuler, R. and Jackson, S. (2005), 'A quarter-century review of human resource management in the US: The growth in importance of the international perspective', *Management Review*, 16, 11–35.
- Scullion, H. (1995) 'International Human Resource Management'. In Storey, J. (ed.) Human Resource Management: A Critical Text. London: Routledge, 352–82.
- Shaw, J.D., Delery, J.E., Jenkins, G.D. and Gupta, N. (1998), 'An organization-level analysis of voluntary and involuntary turnover', *Academy of Management Journal*, 41, 5, 511–525
- Shen, J. and Darby, R. (2006), 'Training and management development in Chinese multinational enterprises', Employee Relations, 28, 4, 342-62.
- Shen, J. and Edwards, V. (2004), 'Recruitment and selection in Chinese MNEs', *International Journal of Human Resource Management*, 15, 814-835.
- Singh, S., Darwish, T., Cristina Costa, A. and Anderson, N. (2012), 'Measuring HRM and organisational performance: Concepts, issues, and framework', *Management Decision*, 50, 4, 1-24.
- Sparrow, P.R., and Hiltrop, J.M. (1994), *European Human Resource Management in Transition*, London: Prentice Hall.
- Sparrow, P.R., Schuler, R.S. and Jackson, S.E. (1994) 'Convergence or divergence: Human resource practices and policies for competitive advantage worldwide', *International Journal of Human Resource Management*, 5, 2, 267–300.
- Szamosi, L.T., Duxbury, L. and Higgins, C. (2004), 'Toward an understanding of people management issues in SMEs: A South-Eastern European perspective', *Education and Training*, 46, 8, 444-465.
- Tan, H. (2001), 'Do training levies work? Malaysia's HRDF and its effects on training and firm-level productivity', Working Paper, World Bank, Washington, DC.
- Vaiman, Vlad and Vance, Charles M. (2008). Smart talent management. Cheltenham, Uk: Edward Elgar.
- Van Smoorenburg, M.S.M. and Van der Velden, R.K.W. (2000), 'The training of school leavers: Complementarity or substitution?', *Economics of Education Review*, 19, 2, 207-217.
- Verburg, R.M. (1998), Human Resource Management: Optimale HRM-praktijken en Con. guraties, Dissertation. Amsterdam: Vrije Universiteit.
- Von Glinow, M.A., Drost, E.A., and Teagarden, M.B. (2002), 'Converging on IHRM best practices: Lessons learnt from a globally distributed consortium on theory and practice', *Asia Pacific Journal of Human Resources*, 40, 1, 146-66.
- Von Glinow, M.A., Huo, Y. and Lowe, P.K. (1999), 'Leadership across the pacific ocean: A transactional comparison', *International Business Review*, 8, 1-15.

- Way, S. (2002), 'High performance work systems and intermediate indicators of firm performance within the US small business sector', *Journal of Management*, 28, 765-785.
- Winterton, J. (2007), 'Training, development and competence' in Boxall, P., Purcell, J. and Wright, P. (eds.), *The Oxford Handbook of Human Resource Management*, Oxford: Oxford University Press.
- Wong, G. and Bimbaum-More, P. (1994), 'Culture, context and structure: A test on Hong Kong banks', *Organisation Studies*, 15, 1, 99-123.
- Wood, S. (1999), 'Human resource management and performance', *International Journal of Management Reviews*, 1, 4, 367 413.
- Wood, S., and de Menezes, L. (2008), 'Comparing perspectives on high involvement management and organizational performance across the British economy', *The International Journal of Human Resource Management*, 19, 4, 639-683.
- Wood, S., Holman, D. and Stride, C. (2006), 'Human resource management and performance in UK call centres', *British Journal of Industrial Relations*, 44, 1, 99-124.
- Wright, P., Szeto, W. F., and Cheng, L. T. W. (2002), 'Guanxi and professional conduct in China: A management development perspective', *International Journal of Human Resource Management*, 13, 1, 156-182.
- Youndt, M.A., Snell, S.A., Dean, J.E., and Lepak, D.P. (1996), 'Human resource management, manufacturing strategy, and firm Performance', *Academy of Management Journal*, 39, 4, 836–865.
- Yuen, E.C. and Kee, H.T. (1993), 'Headquarters, host-culture and organizational influences on HRM policies and practices', *Management International Review*, 33, 361–83.
- Zheng, C., Hyland, P. And Soosay, C. (2007), 'Training practices of multinational companies in Asia', *Journal of European Industrial Training*, 31, 6, 472-494.
- Zheng, C., Morrison, M. and O'Neill, G. (2006), 'An empirical study of high performance HR practices in Chinese SMEs', *International Journal of Human Resource Management*, 17, 10, 1772-1803.

\*\*\*\*

Seri al No.	Recruitment Criteria	Organisatio n	Mean	Mean Difference (Md)	Std. Deviation	t-test statistic	Significance
1	Qualifications 2	3	4	5	6	8	7
1	School and university	DEs	3.99	-0.234	0.903	-1.540	0.063*
	Sensor and university	MNEs	4.22		0.941		
2	Professional qualifications	DEs	3.97	-0.304	0.952	-1.947	0.027**
	Toressional quantications	MNEs	4.27		0.937		
3	Experience in similar jobs	DEs	4.22	-0.260	0.837	-2.159	0.016**
	Zilperieree in chining Jooc	MNEs	4.48		0.644		
4	Wide range of experience	DEs	3.83	-0.139	0.887	-0.920	0.180
	while runge of experience	MNEs	3.97		0.950		
5	Experience in other countries	DEs	3.31	-0.344	1.054	-1.870	0.032**
	zperrenee in outer countres	MNEs	3.65		1.194		
6	Language	DEs	3.78	-0.502	0.915	-3.574	0.000***
	Lunguage	MNEs	4.29		0.750		
7	Presentation	DEs	3.93	-0.211	0.920	-1.342	0.091*
		8MNEs	4.14		0.998		
	Personal Characteristics				T		-
1	Willingness to travel	DEs	3.30	-0.657	1.146	-3.533	0.001***
	while the second	MNEs	3.95		1.099		
2	Devotion to task	DEs	4.16	-0.333	0.981	-2.515	0.007***
		MNEs	4.49		0.644		
3	Self motivation	DEs	4.18	-0.405	0.941	-3.255	0.001***
	Sen motivation	MNEs	4.59		0.586		
4	Potential to grow with the job	DEs	4.00	-0.444	1.028	-2.888	0.002***
	Totential to grow with the job	MNEs	4.44		0.778		
5	Independent judgement	DEs	4.01	-0.433	0.941	-3.195	0.001***
	independent judgement	MNEs	4.44		0.616		
6	Commitment to the company	DEs	4.27	-0.442	0.893	-3.678	0.001***
	communent to the company	MNEs	4.71		0.580		
7	Willingness to learn	DEs	4.31	-0.312	0.862	-2.522	0.007***
	-	MNEs	4.62		0.658		
	Internal / External Recruitmen	nt					
1	Senior manager reporting to	DEs	2.82	0.628	1.327	3.043	0.002***
	Chief Executive	MNEs	2.19		1.134		
2	Junior Manager supervising	DEs	2.77	0.281	1.111	1.615	0.054*
	operatives	MNEs	2.49		0.965		
3	Professional specialist	DEs	3.35	-0.251	1.260	-1.190	0.118
	Fioressional specialist	MNEs	3.60		1.302		

Table-1: Recruitment criteria and internal/external recruitment preferences

1 401	e-2 Freieneu planning for senior po	Januona	1	1	1	1	
N o.	Response to question: 'please indicate your views on managing senior executive succession?'	Organis ation	Mean	Mean Differe nce (Md)	Std. Devia tion	t- statistic	Significance
1	2	3	4	5	6	7	8
1	'External Appointments are, in general, desirable because they bring in new blood'.	DEs MNEs	3.73 2.84	0.886	0.979 1.096	5.216	0.000***
2	'In our case external appointments are undesirable because our company is so large that outsiders cannot understand its complexities'.	DEs MNEs	2.47 2.68	-0.217	1.144 1.468	-0.978	0.165
3	'Competition between internal candidates should be avoided by "preparing" one person well in advance'.	DEs MNEs	2.84 3.27	-0.429	1.249 1.153	-2.148	0.017**
4	'Non-executive directors should play a dominant role in selecting executive successors'.	DEs MNEs	2.68 2.97	-0.286	1.228 1.270	-1.394	0.083*
5	'Appointments are made by the Board of Directors or an equivalent'.	DEs MNEs	3.30 3.14	-0.153	1.366 1.293	0.692	0.245

Table-2 Preferred planning for senior positions

\* Significant at the 0.10 level; \*\*Significant at the 0.05 level; \*\*Significant at the 0.01 level

Table-3: Training methods

Training methods for junior management	Organisation	Mean	Mean Difference (Md)	Std. Deviation	t-statistics	Significance
1. Formal	DEs	4.20	-0.81	0.912	-0.532	0.298
instruction within the company	MNEs	4.29		0.941		
2.Training provided	DEs	3.44	-0.065	1.221	-0.313	0.377
by third party	MNEs	3.51		1.294		
3. Induction by	DEs	3.25	-0.448	1.196	-2.268	0.013**
socialisation	MNEs	3.70		1.200		
4. Learning by	DEs	3.25	-0.004	1.177	-0.020	0.492
doing	MNEs	3.25		1.177		
5. Work placement	DEs	2.91	-0.472	1.100	-2.572	0.006***
with partners	MNEs	3.38		1.128		
6. Buddy system /	DEs	3.17	-0.560	1.224	-3.104	0.001***
mentoring	MNEs	3.73		0.987		

No.	Preferred Criteria	Operation level	Mean	Mean Difference	Std. Deviation	t- statistics	Significance
1	Contribution to profit	DEs MNEs	3.91	-0.027	0.978 0.821	-0.181	0.426
2	Value of output (independent of profit margin)	DEs MNEs	3.94 3.89 3.94	-0.050	0.928 0.821	-0.343	0.366
3	Quality of output	DEs MNEs	4.16 4.29	-0.127	0.829 0.682	-0.995	0.161
4	Keeping within budget	DEs MNEs	3.65 3.86	-0.209	0.885 0.859	-1.452	0.074*
5	Effort (independent of final result)	DEs MNEs	4.06 4.24	-0.181	0.889 0.797	-1.289	0.100*
6	Overall professionalism	DEs MNEs	4.11 4.30	-0.188	0.836 0.733	-1.433	0.077*
7	Deliverables	DEs MNEs	4.14 4.19	-0.054	0.833 0.859	-0.389	0.349
8	Individual competency level (technical)	DEs MNEs	3.95 4.25	-0.299	1.016 0.782	-1.959	0.026**

Table-4: Preferred criteria for internally filling Posts

\* Significant at the 0.10 level; \*\*Significant at the 0.05 level; \*\*\*Significant at the 0.01 level

# Table-5: Turnover rate

	Organisation	Mean	Mean Difference (Md)	Std. Deviation	t-statistics	Sig.
Turnover rate	DEs	2.91	1.084	1.969	3.768	.000***
	MNEs	1.83		1.561		

unent une s	cicculon	, training, a	nu miemai 0			sidering mi	n age			
Operation level	Mean	Mean Difference	Sig.	t- statisti c	Mean	Mean Difference	Sig.	t- stati stic		
ia		Young Firm	ns (0 – 14 yrs)	-	Older firms (15 yrs and above)					
DEs MNEs	4.09 4.59	-0.499	0.015**	-2.500	4.29 4.39	-0.103	0.505	- 0.67 0		
DEs MNEs	3.03 3.70	-0.672	0.033**	-2.191	3.46 3.61	-0.147	0.521	- 0.64 4		
DEs MNEs	3.72 4.33	-0.615	0.007***	-2.791	3.82 4.25	-0.429	0.022**	2.32 6		
acteristics		Young Firm	ns (0 – 14 yrs)		Ol	der firms (15	yrs and above	)		
DEs MNEs	3.22 4.33	-1.115	0.000***	-3.707	3.34 3.67	-0.327	0.162	- 1.40 8		
DEs MNEs	3.91 4.63	-0.723	0.003***	-3.146	4.30 4.39	-0.085	0.615	0.50		
DEs MNEs	3.97 4.52	-0.550	0.016**	-2.495	4.30 4.64	-0.335	0.025**	2.27 3		
DEs MNEs	3.81 4.48	-0.669	0.005***	-2.927	4.11 4.42	-0.310	0.137	- 1.49 9		
DEs MNEs	3.94 4.44	-0.507	0.019**	-2.423	4.05 4.44	-0.391	0.017**	- 2.43 4		
DEs MNEs	4.19 4.59	-0.405	0.074*	-1.819	4.32 4.81	-0.484	0.001***	- 3.59 0		
external		Young Firm	ns (0 – 14 yrs)	Older firms (15 yrs and above)						
DEs MNEs	2.91 1.93	0.980	0.005***	2.906	2.77 2.39	0.379	0.141	1.48 4		
		Young Firm	ns (0 – 14 yrs)		Ol	der firms (15	yrs and above	)		
DEs MNEs	3.09 4.00	-0.906	0.006***	-2.845	3.34 3.47	-0.133	0.597	0.53		
DEs MNEs	2.66 3.33	-0.677	0.024**	-2.318	3.05 3.42	-0.363	0.127	- 1.54 0		
DEs MNEs	2.94 3.67	-0.729	0.027**	-2.273	3.30 3.78	-0.474	0.040**	2.08		
tunity		Young Firm	ns (0 – 14 yrs)	•	Ol	der firms (15	yrs and above			
DEs MNEs	3.72 4.26	-0.541	0.038**	-2.129	4.09 4.25	-0.161	0.387	- 0.87 0		
		Young Firm	ns (0 – 14 yrs)		Older firms (15 yrs and above)					
DEs MNEs	3.69 2.59	1.095	0.000***	3.744	3.75 3.03	0.722	0.001***	3.51 1		
	Operation         level         DEs         MNEs         DEs <td>Operation levelMeania-DEs MNEs4.09 4.59DEs MNEs3.03 3.70DEs MNEs3.72 4.33DEs DEs MNEs3.72 4.33acteristics-DEs MNEs3.22 4.33DEs DEs MNEs3.91 4.63DEs MNEs3.91 4.63DEs MNEs3.91 4.63DEs MNEs3.91 4.63DEs MNEs3.91 4.52DEs MNEs3.91 4.52DEs MNEs3.94 4.44DEs MNEs4.19 4.59external DEs MNEs-DEs MNEs3.09 4.00DEs MNEs2.91 4.00DEs MNEs3.09 4.00DEs MNEs2.94 3.67tunity DEs MNEs-DEs MNEs3.72 4.26DEs MNEs3.69</td> <td>Operation levelMeanMean DifferenceiaYoung FirmDEs<math>4.09</math> MNEs<math>4.59</math>DEs<math>3.03</math> MNEs<math>-0.499</math>DEs<math>3.70</math><math>-0.672</math>DEs<math>3.70</math><math>-0.672</math>DEs<math>3.72</math> MNEs<math>4.33</math>MNEs<math>4.33</math><math>-0.615</math>acteristicsYoung FirmDEs<math>3.22</math> MNEs<math>4.63</math>MNEs<math>4.63</math><math>-0.723</math>DEs<math>3.91</math> MNEs<math>4.63</math>MNEs<math>4.63</math><math>-0.723</math>DEs<math>3.91</math> MNEs<math>4.63</math>MNEs<math>4.44</math><math>-0.507</math>DEs<math>3.94</math> MNEs<math>4.44</math>MNEs<math>4.44</math><math>0.669</math>DEs<math>3.94</math> MNEs<math>4.00</math>DEs<math>3.94</math> MNEs<math>4.63</math>MNEs<math>4.59</math><math>-0.405</math>externalYoung FirmDEs<math>3.09</math> MNEs<math>3.67</math>DEs<math>3.67</math> A.00<math>-0.906</math>DEs<math>2.94</math> MNEs<math>-0.729</math>tunityYoung FirmDEs<math>3.67</math> A.00<math>-0.729</math>tunityYoung FirmDEs<math>3.69</math> A.00<math>-0.541</math></td> <td>Operation level         Mean         Mean Difference         Sig.           ia         Young Firms (0 – 14 yrs)         <math>(-14 yrs)</math>           DEs         4.09         <math>-0.499</math> <math>0.015^{**}</math>           DEs         3.03         <math>-0.672</math> <math>0.033^{**}</math>           DEs         3.72         <math>-0.615</math> <math>0.007^{***}</math>           DEs         3.72         <math>0.033^{**}</math> <math>0.007^{***}</math>           Acteristics         Young Firms (0 – 14 yrs)         <math>0.007^{***}</math>           DEs         3.22         <math>0.007^{***}</math> <math>0.007^{***}</math>           DEs         3.91         <math>0.003^{***}</math> <math>0.000^{***}</math>           DEs         3.91         <math>0.003^{***}</math> <math>0.003^{***}</math>           DEs         3.91         <math>0.003^{***}</math> <math>0.003^{***}</math>           DEs         3.91         <math>0.003^{***}</math> <math>0.003^{***}</math>           DEs         3.91         <math>0.005^{***}</math> <math>0.005^{***}</math>           DEs         3.94         <math>0.005^{***}</math> <math>0.074^{**}</math>           DEs         4.19         <math>0.980</math> <math>0.005^{***}</math>           DEs         3.09         <math>0.980</math> <math>0.005^{***}</math>           DEs         3.09         <math>0.980</math> <math>0.006^{***}</math></td> <td>Operation level         Mean         Mean Difference         Sig.         t- statisti c           DEs MNES         4.09 4.59         -0.499         0.015**         -2.500           DEs MNES         3.03 3.70         -0.672         0.033**         -2.191           DEs MNES         3.70         -0.615         0.007***         -2.791           acteristics         Young Firms (0 - 14 yrs)         -2.791           acteristics         Young Firms (0 - 14 yrs)         -2.791           DEs MNES         3.22 4.33         -0.615         0.007***         -2.791           acteristics         Young Firms (0 - 14 yrs)         -2.791           DEs MNEs         3.91 4.63         -0.723         0.003***         -3.707           DEs MNEs         3.91 4.63         -0.723         0.003***         -2.495           DEs MNEs         3.91 4.63         -0.550         0.016**         -2.495           DEs MNEs         3.94 4.44         -0.507         0.019**         -2.495           DEs MNEs         2.91 4.99         -0.405         0.074*         -1.819           external         Young Firms (0 - 14 yrs)         -2.906         -2.906           DEs MNEs         3.09 4.00         -0.906         0.006</td> <td><math display="block">\begin{array}{c c c c c c c c c c c c c c c c c c c </math></td> <td><math display="block">\begin{array}{ c c c c c c c c c c c c c c c c c c c</math></td> <td><math display="block">\begin{array}{ c c c c c c } \hline \text{Mean} &amp; \begin{array}{ c c c c c } \hline \text{Mean} &amp; \begin{array}{ c c c c } \hline \text{Mean} &amp; \begin{array}{ c } \hline \ \text{Mean} &amp; \begin{array}{ c } \hline \ \text{Mean} &amp; \begin{array}{ c } \hline \ \text{Mean} &amp; \begin{array}{ c } \hline \hline \ \text{Mean} &amp; \begin{array}{ c } \hline \hline \ \ \text{Mean} &amp; \begin{array}{ c } \hline \end{array} \end{array} \end{array} \end{array} \end{array} \end{array} } \end{array} \end{array} } \end{array} \right </math></td>	Operation levelMeania-DEs MNEs4.09 4.59DEs MNEs3.03 3.70DEs MNEs3.72 4.33DEs DEs MNEs3.72 4.33acteristics-DEs MNEs3.22 4.33DEs DEs MNEs3.91 4.63DEs MNEs3.91 4.63DEs MNEs3.91 4.63DEs MNEs3.91 4.63DEs MNEs3.91 4.52DEs MNEs3.91 4.52DEs MNEs3.94 4.44DEs MNEs4.19 4.59external DEs MNEs-DEs MNEs3.09 4.00DEs MNEs2.91 4.00DEs MNEs3.09 4.00DEs MNEs2.94 3.67tunity DEs MNEs-DEs MNEs3.72 4.26DEs MNEs3.69	Operation levelMeanMean DifferenceiaYoung FirmDEs $4.09$ MNEs $4.59$ DEs $3.03$ MNEs $-0.499$ DEs $3.70$ $-0.672$ DEs $3.70$ $-0.672$ DEs $3.72$ MNEs $4.33$ MNEs $4.33$ $-0.615$ acteristicsYoung FirmDEs $3.22$ MNEs $4.63$ MNEs $4.63$ $-0.723$ DEs $3.91$ MNEs $4.63$ MNEs $4.63$ $-0.723$ DEs $3.91$ MNEs $4.63$ MNEs $4.44$ $-0.507$ DEs $3.94$ MNEs $4.44$ MNEs $4.44$ $0.669$ DEs $3.94$ MNEs $4.00$ DEs $3.94$ MNEs $4.63$ MNEs $4.59$ $-0.405$ externalYoung FirmDEs $3.09$ MNEs $3.67$ DEs $3.67$ A.00 $-0.906$ DEs $2.94$ MNEs $-0.729$ tunityYoung FirmDEs $3.67$ A.00 $-0.729$ tunityYoung FirmDEs $3.69$ A.00 $-0.541$	Operation level         Mean         Mean Difference         Sig.           ia         Young Firms (0 – 14 yrs) $(-14 yrs)$ DEs         4.09 $-0.499$ $0.015^{**}$ DEs         3.03 $-0.672$ $0.033^{**}$ DEs         3.72 $-0.615$ $0.007^{***}$ DEs         3.72 $0.033^{**}$ $0.007^{***}$ Acteristics         Young Firms (0 – 14 yrs) $0.007^{***}$ DEs         3.22 $0.007^{***}$ $0.007^{***}$ DEs         3.91 $0.003^{***}$ $0.000^{***}$ DEs         3.91 $0.003^{***}$ $0.003^{***}$ DEs         3.91 $0.003^{***}$ $0.003^{***}$ DEs         3.91 $0.003^{***}$ $0.003^{***}$ DEs         3.91 $0.005^{***}$ $0.005^{***}$ DEs         3.94 $0.005^{***}$ $0.074^{**}$ DEs         4.19 $0.980$ $0.005^{***}$ DEs         3.09 $0.980$ $0.005^{***}$ DEs         3.09 $0.980$ $0.006^{***}$	Operation level         Mean         Mean Difference         Sig.         t- statisti c           DEs MNES         4.09 4.59         -0.499         0.015**         -2.500           DEs MNES         3.03 3.70         -0.672         0.033**         -2.191           DEs MNES         3.70         -0.615         0.007***         -2.791           acteristics         Young Firms (0 - 14 yrs)         -2.791           acteristics         Young Firms (0 - 14 yrs)         -2.791           DEs MNES         3.22 4.33         -0.615         0.007***         -2.791           acteristics         Young Firms (0 - 14 yrs)         -2.791           DEs MNEs         3.91 4.63         -0.723         0.003***         -3.707           DEs MNEs         3.91 4.63         -0.723         0.003***         -2.495           DEs MNEs         3.91 4.63         -0.550         0.016**         -2.495           DEs MNEs         3.94 4.44         -0.507         0.019**         -2.495           DEs MNEs         2.91 4.99         -0.405         0.074*         -1.819           external         Young Firms (0 - 14 yrs)         -2.906         -2.906           DEs MNEs         3.09 4.00         -0.906         0.006	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	$\begin{array}{ c c c c c c } \hline \text{Mean} & \begin{array}{ c c c c c } \hline \text{Mean} & \begin{array}{ c c c c } \hline \text{Mean} & \begin{array}{ c } \hline \ \text{Mean} & \begin{array}{ c } \hline \ \text{Mean} & \begin{array}{ c } \hline \ \text{Mean} & \begin{array}{ c } \hline \hline \ \text{Mean} & \begin{array}{ c } \hline \hline \ \ \text{Mean} & \begin{array}{ c } \hline \end{array} \end{array} \end{array} \end{array} \end{array} \end{array} } \end{array} \end{array} } \end{array} \right $		

Table-6: Recruitment and selection, training, and internal opportunities considering firm age

	Operation level	Mean	Mean Difference	Sig.	t- statistic	Mean	Mean Difference	Sig.	t-statistic	
Routine Crite		S	mall Firms (0	- 50 employee		I		employees and	above)	
Experience	DEs	4.14	man Pirms (0	- 50 employed	-3)	4.32	arge mins (51	employees and		
in similar job	MNEs	4.55	-0.405	0.014**	-2.505	4.40	-0.084	0.662	-0.440	
Experience in other countries	DEs MNEs	3.32 3.36	-0.044	0.870	-0.164	3.29 3.97	-0.677	0.007***	-2.799	
Language	DEs MNEs	3.88 4.33	-0.453	0.009***	-2.664	3.66 4.23	-0.575	0.016**	-2.473	
Personal Cha	racteristics	S	mall Firms (0	- 50 employee	es)	L	arge firms (51	employees and	above)	
Travel	DEs MNEs	3.26 3.73	-0.467	0.075*	-1.803	3.34 4.20	-0.858	0.002***	-3.245	
Dedication to task	DEs MNEs	4.10 4.55	-0.445	0.014**	-2.517	4.24 4.43	-0.196	0.350	-0.941	
Self motivation	DEs MNEs	4.12 4.61	-0.486	0.003***	-3.051	4.26 4.57	-0.304	0.132	-1.524	
Grow with the job	DEs MNEs	3.88 4.42	-0.544	0.015**	-2.483	4.16 4.47	-0.309	0.132	-1.526	
Independent judgement	DEs MNEs	3.90 4.48	-0.585	0.003***	-3.116	4.16 4.40	-0.242	0.195	-1.311	
Commitmen t	DEs MNEs	4.20 4.79	-0.588	0.000***	-3.840	4.37 4.63	-0.265	0.178	-1.363	
Willingness to learn	DEs MNEs	4.30 4.70	-0.397	0.019**	-2.396	4.32 4.53	-0.218	0.278	-1.095	
Internal recruitment	external	S	mall Firms (0	- 50 employee	es)	Large firms (51 employees and above)				
Seniors manager to chief executive	DEs MNEs	2.82 2.24	0.578	0.047**	2.022	2.82 2.13	0.682	0.027**	2.257	
Quality of domestic managers	DEs MNEs	2.98 2.73	0.253	0.200	1.293	3.11 2.33	0.772	0.001***	3.486	
Training		S	mall Firms (0	- 50 employee	es)	Large firms (51 employees and above)				
Induction by socialisation / imitation	DEs MNEs	3.22 3.52	-0.295	0.264	-1.125	3.29 3.90	-0.611	0.047**	-2.026	
Placement	DEs MNEs	3.12 3.67	-0.547	0.025**	-2.279	2.63 3.07	-0.435	0.112	-1.611	
Buddy system	DEs MNEs	3.30 3.88	-0.579	0.020**	-2.370	3.00 3.57	-0.567	0.053*	-1.972	
Senior succes		S	mall Firms (0	- 50 employee	es)	L	arge firms (51	employees and	above)	
External appointment	DEs MNEs	3.76 2.91	0.851	0.000***	3.643	3.68 2.77	0.918	0.001***	3.659	
Appointed by Board of Directors	DEs MNEs	3.14 3.61	-0.466	0.113	-1.604	3.50 2.63	0.867	0.008***	2.718	

Table-7: Recruitment and selection, training, and internal opportunities by firm size

Serial No.	HR practices as a whole	Organisation	Mean	Mean Difference (Md)	Std. Deviation	t-test statistic	Significance
1	2	3	4	5	6	8	7
	HR Practices						
1	Recruitment and Selection	DEs MNEs	79.20 83.46	-4.256	10.492 8.846	-2.621	0.010**
2	Training	DEs MNEs	20.23 21.86	-1.630	3.900 3.573	-2.621	0.010**
3	Internal Career Opportunities	DEs MNEs	31.86 33.00	-1.136	5.222 4.429	-1.403	0.163
<b>The I</b>	Internal/Horizontal Fit of E	IR Practices (HF DEs MNEs	<b>R Practices in</b> 1624.32 1841.68	tegrated and c	oherent as a s 451.608 433.027	system/bundle	of Practices)
2	Selection with TrainingRecruitment andSelection with InternalCareer Opportunities	DEs MNEs	2560.47 2781.19	-220.520	653.384 599.368	-2.116	0.036**
3	Recruitment and Selection with Turnover	DEs MNEs	224.43 148.54	75.892	150.201 121.009	3.433	0.001***
4	Training with Internal Career Opportunities	DEs MNEs	654.44 729.35	-74.906	197.172 187.929	-2.347	0.020**
4 5				-74.906 17.602		-2.347 3.092	0.020**

Table-8 HR practices as an integrated and coherent system